Author's response to reviews

Title: Mental health of refugees from Germany following state-sponsored repatriation

Authors:

Ulrike von Lersner (ulrike.von.lersner@psychologie.hu-berlin.de)
Thomas Elbert (thomas.elbert@uni-konstanz.de)
Frank Neuner (frank.neuner@uni-konstanz.de)

Version: 6 Date: 21 October 2008

Author's response to reviews:

I completed the formatting changes that have been requested. One request was to ensure that the title is identical in the manuscript and the submission system. I changed the title after the first review as this was requested by one of the reviewers. This might explain why it is not identical anymore in the submission system.